

# **Geophysical Research Letters**<sup>•</sup>

# **RESEARCH LETTER**

10.1029/2021GL095717

# **Key Points:**

- Responses of total ozone deposition to heat and dry anomalies vary considerably from site to site
- Non-stomatal deposition increases significantly during hot days in all three sites considered
- Current big-leaf parameterizations largely fail to capture the response mainly because of non-stomatal deposition

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

J. A. Geddes, jgeddes@bu.edu

#### Citation:

Wong, A. Y. H., Geddes, J. A., Ducker, J. A., Holmes, C. D., Fares, S., Goldstein, A. H., et al. (2022). New evidence for the importance of non-stomatal pathways in ozone deposition during extreme heat and dry anomalies. *Geophysical Research Letters*, 49, e2021GL095717. https://doi.org/10.1029/2021GL095717

Received 13 SEP 2021 Accepted 28 MAR 2022

#### **Author Contributions:**

Conceptualization: A. Y. H. Wong, J. A. Geddes Data curation: S. Fares, A. H. Goldstein. I. Mammarella, J. W. Munger Formal analysis: A. Y. H. Wong Funding acquisition: J. A. Geddes Investigation: A. Y. H. Wong, J. A. Geddes Methodology: A. Y. H. Wong, J. A. Geddes Project Administration: J. A. Geddes Software: A. Y. H. Wong, J. A. Ducker, C. D. Holmes Supervision: J. A. Geddes Validation: A. Y. H. Wong Visualization: A. Y. H. Wong Writing - original draft: A. Y. H. Wong Writing - review & editing: A. Y. H. Wong, J. A. Geddes, J. A. Ducker, C. D. Holmes, S. Fares, A. H. Goldstein, I. Mammarella, J. W. Munger

© 2022. American Geophysical Union. All Rights Reserved.

# New Evidence for the Importance of Non-Stomatal Pathways in Ozone Deposition During Extreme Heat and Dry Anomalies

A. Y. H. Wong<sup>1</sup>, J. A. Geddes<sup>1</sup>, J. A. Ducker<sup>2</sup>, C. D. Holmes<sup>2</sup>, S. Fares<sup>3,4</sup>, A. H. Goldstein<sup>5,6</sup>, I. Mammarella<sup>7</sup>, and J. W. Munger<sup>8</sup>

<sup>1</sup>Department of Earth and Environment, Boston University, Boston, MA, USA, <sup>2</sup>Department of Earth, Ocean and Atmospheric Science, Florida State University, Tallahassee, FL, USA, <sup>3</sup>Research Centre of Forestry and Wood, Council of Agricultural Research and Economics, Rome, Italy, <sup>4</sup>National Research Council, Institute of Bioeconomy, Rome, Italy, <sup>5</sup>Department of Environmental Science, Policy, and Management, University of California, Berkeley, CA, USA, <sup>6</sup>Department of Civil and Environmental Engineering, University of California, Berkeley, CA, USA, <sup>7</sup>Institute for Atmospheric and Earth System Research/Physics, University of Helsinki, Finland, <sup>8</sup>School of Engineering and Applied Science and Department of Earth and Planetary Sciences, Harvard University, Cambridge, MA, USA

**Abstract** Dry deposition could partially explain the observed response in ambient ozone to extreme hot and dry episodes. We examine the response of ozone deposition to heat and dry anomalies using three long-term co-located ecosystem-scale carbon dioxide, water vapor and ozone flux measurement records. We find that, as expected, canopy stomatal conductance generally decreases during days with dry air or soil. However, during hot days, concurrent increases in non-stomatal conductance are inferred at all three sites, which may be related to several temperature-sensitive processes not represented in the current generation of big-leaf models. This may offset the reduction in stomatal conductance, leading to smaller net reduction, or even net increase, in total deposition velocity. We find the response of deposition velocity to soil dryness may be related to its impact on photosynthetic activity, though considerable variability exists. Our findings emphasize the need for better understanding and representation of non-stomatal ozone deposition.

**Plain Language Summary** Ozone is an important air pollutant that can threaten both human and plant health. Removal of ozone from the atmosphere may be reduced during extremely hot or dry events due to how plants respond to such environmental conditions (governed by stomatal or non-stomatal processes separately). Using long-term observations at three different sites, we find that non-stomatal uptake generally increases on hot days, which can offset a reduction in stomatal uptake that is expected under the same conditions. The response to soil dryness is more complicated, but potentially related to responses in photosynthetic activity. Current models of on how ozone deposition affects surface ozone concentrations during hot and dry episodes are inaccurate because of their inability to represent non-stomatal responses.

# 1. Introduction

Surface ozone  $(O_3)$  is an important air pollutant with adverse effects on human health (Jerrett et al., 2009), and ecosystem productivity (Ainsworth et al., 2012; Tai et al., 2014; Wittig et al., 2009). One of its major atmospheric sinks is through dry deposition (Wild, 2007), referring to the removal of atmospheric trace chemicals by turbulent transport to the Earth surface (Wesely & Hicks, 2000). Terrestrial ecosystems are efficient sinks of surface  $O_3$  because of both stomatal uptake and non-stomatal processes (e.g., uptake on cuticles and soil, or in-canopy gas-phase chemistry) (Fowler et al., 2009). Though process-level knowledge remains incomplete (Clifton et al., 2020), observational evidence indicates that  $O_3$  dry deposition over terrestrial ecosystems exhibits strong variability from diurnal to interannual timescales (O. E. Clifton et al., 2017, 2019; Fares et al., 2010, 2012, 2014; Rannik et al., 2012; Ronan et al., 2020; Wong et al., 2019; Zona et al., 2014). Predictions of surface  $O_3$  will benefit from a better understanding of the temporal dynamics of its dry deposition.

Dry deposition is expected to affect surface  $O_3$  levels during hot and dry episodes. For example, the particularly hot and dry conditions in 2006 may have significantly reduced dry deposition, and therefore enhanced surface  $O_3$  concentrations, over the United Kingdom that summer (Emberson et al., 2013). M. Lin et al. (2019) also argue that drought and heat-induced reduction in dry deposition contributes to the high surface  $O_3$  over the central and eastern United States in summer 2012.

This literature generally relies on the assumption that stomatal conductance, and therefore  $O_3$  uptake, is suppressed by heat and dryness. Less attention has been given to how non-stomatal  $O_3$  uptake may also change under such conditions. Low relative humidity may reduce cuticular  $O_3$  uptake (Sun et al., 2016; Zhang et al., 2002), or high temperatures may promote cuticular  $O_3$  uptake through faster surface reactions (Cape et al., 2009). In some forests where direct ozonolysis by biogenic volatile organic compounds (BVOC) plays an important role in  $O_3$ uptake, the inferred non-stomatal uptake could increase as a function of temperature (Kurpius & Goldstein, 2003; Vermeuel et al., 2021; Wolfe et al., 2011). Dry soil may also promote  $O_3$  uptake to soil (Mészáros et al., 2009; Stella et al., 2019; Stella, Loubet, et al., 2011). With these potentially competing pathways, the response of non-stomatal  $O_3$  uptake to heat and dry anomalies is highly uncertain (e.g., Q. Li et al., 2019).

Extreme dryness and heat are expected to become more frequent and severe (Dai & Zhao, 2017; Meehl & Tebaldi, 2004; Perkins et al., 2012; Samaniego et al., 2018). Together with the empirical evidence that the above conditions could lead to increase in  $O_3$  levels at the surface with a concomitant additional public health burden (Filleul et al., 2006), it is important to mechanistically understand  $O_3$  dry deposition to correctly predict the changes in surface  $O_3$  and related risks. Characterizing ecosystem responses to extreme events typically requires analysis of long-term monitoring data (Chu et al., 2017; Zscheischler et al., 2014) that enables comparisons across similar seasonal and phenological conditions.

We leverage multi-year  $O_3$  flux measurements at several sites to explore the response of  $O_3$  dry deposition under extreme dryness and heat. We select sites with co-located sensible heat, latent heat and  $CO_2$  flux measurements, so that we can partition total  $O_3$  deposition into stomatal and non-stomatal pathways (Fares et al., 2012; Gerosa et al., 2005; Hogg et al., 2007; Stella, Personne, et al., 2011), and characterize associated ecosystem stress. This approach allows us the investigate the following questions:

- 1. How does total, stomatal and non-stomatal O3 deposition change under heat and dry anomalies?
- 2. What are the plausible mechanisms and their potential ties to meteorology and ecosystem stress behind such changes?
- 3. Can big-leaf parameterizations used by regional and global models capture the variability in O<sub>3</sub> deposition during heat and dry episodes?

#### 2. Datasets and Method

We use long-term eddy covariance (EC) measurements of sensible heat (*H*), latent heat (*LE*),  $O_3$ , and  $CO_2$  fluxes, and relevant auxiliary meteorological variables (e.g., air temperature, humidity, and radiation) from three sites: (a) Hyytiälä Forest (*Hyy*), Finland (Keronen et al., 2003; Mammarella et al., 2007); (b) Harvard Forest (*Ha*), Massachusetts, USA (Munger et al., 1996); and (c) Blodgett Forest (*Blo*), California, USA (Fares et al., 2010). *Hyy* and *Blo* are characterized as needleleaf forests, while *Ha* is characterized as deciduous forest with scattered stands of needleleaf species. We include additional details of each site in Table S1 in Supporting Information S1.

Canopy conductance of  $O_3(g_{c,O3})$ , representing the strength of the  $O_3$  sink to the surface, is calculated by:

$$g_{c,O_3} = \left(v_{d,O_3}^{-1} - r_a - r_{b,O_3}\right)^{-1}$$
(1)

where  $v_d$  is O<sub>3</sub> deposition velocity (inferred from the O<sub>3</sub> flux and concentration measurements),  $r_a$  is the aerodynamic resistance (inferred based on widely accepted Monin-Obukhov similarity theory (Foken, 2006; Monin & Obukhov, 1954)), and  $r_b$  is the laminar boundary-layer resistance (calculated based on the formula proposed by Wesely and Hicks (1977)). We reject observations with low turbulence (friction velocity <0.1 m s<sup>-1</sup>) as  $v_d$  is more likely to be controlled by  $r_a$  rather than  $g_{c,O3}$ , and often characterized by large random error (Rannik et al., 2012). This filter removes less than 1.2% of the daytime hourly observations (Table S1 in Supporting Information S1).

We apply three different strategies to infer canopy stomatal conductance of water vapor  $(g_{s,w})$ : (a) The evaporative-resistive form of Penman-Monteith (PM) equation (Gerosa et al., 2007; Monteith, 1965) with the Nelson et al. (2018) machine learning-based method to estimate ecosystem transpiration, (b) the PM equation with a simpler set of assumptions in evapotranspiration (*ET*) partitioning, and (c) a *GPP*-based estimate (Y. S. Lin et al., 2015; Medlyn et al., 2011). Detail descriptions of the methods, and the reasons supporting the use of



method 1 are given in Text S1 in Supporting Information S1. Stomatal conductance of  $O_3(g_{s,O3})$  is then scaled from  $g_{s,w}$  by the relative diffusivity between H<sub>2</sub>O and O<sub>3</sub> molecules (Wesely, 1989):

$$g_{s,O_3} = \frac{g_{s,w}}{1.6}$$
(2)

The residual of  $v_d$  is then partitioned to estimate an apparent (or inferred) non-stomatal conductance  $(g_{n_s,Q_s})$ :

۶

$$g_{ns,O_3} = g_{c,O_3} - g_{s,O_3} \tag{3}$$

Regional and global models tend to use big-leaf parameterizations of  $v_d$  (Hardacre et al., 2015; Pleim & Ran, 2011; Simpson et al., 2012). To investigate their performance, we model  $v_d$ ,  $g_{s,O3}$ , and  $g_{ns,O3}$  with two widely-used big-leaf parameterizations: the Wesely scheme (Wesely, 1989), and the Zhang scheme (Zhang et al., 2003). Details of each are given in Table S2 in Supporting Information S1.

We focus on summer daytime observations (9a.m.–3p.m. local time) when  $O_3$  deposition is highest and boundary-layer turbulence is most developed (Freire et al., 2017). The definition of summertime for each site is taken from previous studies (Clifton et al., 2017; Fares et al., 2010; Rannik et al., 2012) (see Table S1 in Supporting Information S1). Daily average observed and modeled  $v_d$ ,  $g_{s,O3}$ ,  $g_{ns,O3}$ , air temperature (*T*), vapor pressure deficit (*VPD*), incoming solar radiation (*SW<sub>in</sub>*), soil water content (*SWC*) and gross primary productivity (*GPP*, "GPP\_ NT\_VUT\_REF" from FLUXNET 2015 (Pastorello et al., 2020)) are computed for days with no more than 2 hours of missing daytime  $v_d$  observations.

Finally, we define days with 10% highest daytime average *T*, *VPD*, and 10% lowest daytime average *SWC* as "anomalously" hot (high *T*), dry air (high *VPD*), and dry soil (low *SWC*) days respectively. The choice of 90th percentile provides reasonable sample size and corresponds to accepted definitions of anomalous events (Perkins & Alexander, 2013; Perkins et al., 2012). Other days are labeled as "normal".

#### 3. Results and Discussions

Table 1 compares the mean and standard deviation of daytime average VPD, T, SWC,  $SW_{in}$ , and GPP during "anomalous" days with rest of the sample population at each site. The mean daytime T of 10% hottest days is 5.7°C–7.7°C higher than the average of other summer days. As VPD is partly dependent on temperature through the strong relationship between saturated vapor pressure and air temperature (Alduchov & Eskridge, 1996), high T naturally increases VPD. At Hyy and Blo, many high T days (~30) overlap with high VPD days. At Ha, this co-occurrence is less common (14 days). Still, we find enough distinction between the populations and ecological impacts of high T and high VPD days that they can be studied separately. GPP shows slight increases or no changes during both hot and high VPD days at Hyy and Ha, while at Blo hot days and high VPD days have opposing responses on GPP (+4% and -20%, respectively). At all three sites, dry soil days have little overlap with either high T or high VPD (2–9 days), providing a mostly distinct condition to study. Dry soil conditions are associated with suppressed GPP, though to varying degrees across all sites (-14% in Hyy, -59% in Blo and -24% in Ha).

Figure 1 summarizes the  $v_d$ ,  $g_{s,O3}$ , and  $g_{ns,O3}$  inferred from observations under normal and anomalous conditions obtained from the Nelson et al. (2018) transpiration scheme, and compares them with predictions from the two big-leaf models. We calculate the significance of differences in response (medians) between the normal and anomalous days with a Wilcoxon Rank-Sum Test (Wilcoxon, 1945). Both the observed and modeled responses of  $v_d$  to anomalous conditions vary considerably across sites.

#### 3.1. Heat and High VPD Anomalies

During the 10% hottest days, observed  $v_d$  is -0.08 cm s<sup>-1</sup> (14%) lower at Hyy, but +0.10 cm s<sup>-1</sup> (16%) higher over *Blo*. At *Ha*  $v_d$  is slightly reduced but the difference is not statistically significant. We find that the inferred  $g_{s,O3}$  shows strong declines at Hyy (-0.15 cm s<sup>-1</sup>, 37%), but does not decline significantly at *Blo* and *Ha*. At all three sites, the inferred  $g_{ns,O3}$  is significantly higher during hot days (+0.10 to +0.18 cm s<sup>-1</sup>). The overall  $v_d$  response to extreme heat is therefore determined by whether the reduction in  $g_{s,O3}$  can compensate for the increase in  $g_{ns,O3}$ .



## Table 1

Average Daytime (9a.m.–3p.m.) Mean VPD, T, SWC, SW<sub>in</sub> and GPP From Days With and Without Anomalous Conditions for All Three Sites

	T (°C)	VPD (kPa)	SWC (%)	$SW_{in} (W m^{-2})$	GPP ( $\mu$ molC m <sup>-2</sup> s <sup>-1</sup> )
Hyytiälä					
$T \ge 90\%$ ile	$24.9 \pm 1.5^*$	$1.60\pm0.42^*$	$25.6 \pm 4.8 *$	$546 \pm 92^*$	$8.43 \pm 1.66$
T < 90%ile	$17.2 \pm 2.9$	$0.76 \pm 0.39$	$27.6\pm6.9$	$392 \pm 157$	$7.84 \pm 2.03$
$VPD \ge 90\%$ ile	$23.6 \pm 2.6 *$	$1.75\pm0.28^*$	$26.6\pm5.5$	$598 \pm 57*$	$8.58 \pm 1.64*$
<i>VPD</i> < 90%ile	$17.4 \pm 3.2$	$0.75 \pm 0.37$	$27.5 \pm 6.8$	$387 \pm 152$	$7.81 \pm 2.02$
<i>SWC</i> < 10%ile	$18.5 \pm 4.0$	$0.85 \pm 0.58$	$17.9 \pm 1.2*$	$415 \pm 158$	$6.91 \pm 1.93^*$
$SWC \ge 10\%$ ile	$18.1 \pm 3.7$	$0.86 \pm 0.47$	$28.4 \pm 6.2$	377 ± 153	$8.05 \pm 1.97$
Blodgett					
$T \ge 90\%$ ile	$28.9 \pm 1.0^*$	$3.00\pm0.29^*$	$16.6 \pm 2.0$	$780 \pm 114$	$10.7 \pm 5.2$
T < 90%ile	$23.2 \pm 3.5$	$2.02 \pm 0.58$	$16.2 \pm 3.0$	$778 \pm 106$	$10.3 \pm 4.3$
$VPD \ge 90\%$ ile	$28.5 \pm 1.3^*$	$3.10\pm0.18^*$	$15.6 \pm 1.7*$	$768 \pm 102$	$8.45 \pm 4.30^{*}$
<i>VPD</i> < 90%ile	$23.2 \pm 3.6$	$2.00 \pm 0.57$	$16.4 \pm 3.0$	779 ± 114	$10.6 \pm 5.16$
<i>SWC</i> < 10%ile	$22.9 \pm 5.6$	$2.19 \pm 0.82$	$13.7\pm0.2*$	$725 \pm 127*$	$4.32 \pm 2.56^{*}$
$SWC \ge 10\%$ ile	$24.0\pm3.3$	$2.11 \pm 0.61$	$16.6 \pm 2.9$	809 ± 98	$10.6 \pm 4.29$
Harvard					
$T \ge 90\%$ ile	$27.3 \pm 1.3 *$	$1.31\pm0.40^*$	$20.9\pm8.3^*$	$669 \pm 92*$	$22.5 \pm 4.5$
T < 90%ile	$20.6 \pm 3.1$	$0.82 \pm 0.46$	$27.6 \pm 9.9$	$547 \pm 209$	$20.4 \pm 5.2$
$VPD \ge 90\%$ ile	$24.9 \pm 2.1 *$	$1.66\pm0.17^*$	$21.6 \pm 8.7 *$	$727* \pm 84$	$20.6 \pm 5.2$
<i>VPD</i> < 90%ile	$20.9 \pm 3.5$	$0.78 \pm 0.41$	$27.7 \pm 9.9$	$540 \pm 204$	$20.6 \pm 4.6$
<i>SWC</i> < 10%ile	$21.8 \pm 3.9$	$1.08 \pm 0.53$	$11.4 \pm 1.9^{*}$	$589 \pm 207$	$16.2\pm4.6^*$
$SWC \ge 10\%$ ile	$21.3 \pm 3.5$	$0.86 \pm 0.48$	28.7 ± 8.9	543 ± 196	$21.4 \pm 5.0$

*Note.* Asterisks indicate statistically significant (p < 0.01) difference between extreme and non-extreme days.

We find that neither the Wesely nor Zhang parameterization captures the increases in inferred  $g_{ns,O3}$ , and therefore do not correctly capture the observed responses of  $v_d$  to extreme heat. At *Hyy*, competing errors in the Zhang parameterization (overpredicting the reduction in  $g_{s,O3}$  and underpredicting the reduction in  $g_{ns,O3}$ ) result in an overall reduction in  $v_d$  that is comparable to that inferred by observations. Still, the Zhang parameterization tends to capture the reduction in  $g_{s,O3}$  better than the Wesely parameterization. This is not surprising, since the former includes land cover-specific stomatal response to *T* and *VPD*, while  $g_{s,O3}$  in the Wesely parameterization lacks any *VPD* dependence (and has fixed optimal temperature for stomatal opening irrespective of plant type and climate).

We find that high VPD generally leads to stronger reductions in inferred  $g_{s,03}$  at all sites, with either weaker (*Ha*) or no increases (*Hyy* and *Blo*) in  $g_{ns,03}$ . At *Hyy*, the  $v_d$  change with high VPD (-0.09 cm s<sup>-1</sup>, -15%) is comparable to that during heat anomalies. In contrast, high VPD at *Blo* reduces  $v_d$  by -0.06 cm s<sup>-1</sup> (-10%, not significant at 95% level), attributable to the stronger reduction in  $g_{s,03}$  (-0.11 cm s<sup>-1</sup>) and the lack of response in the inferred  $g_{ns,03}$ . At *Ha*, the reduction in inferred  $g_{s,03}$  (-0.11 cm s<sup>-1</sup>) and increase in inferred  $g_{ns,03}$  (+0.11 cm s<sup>-1</sup>) largely offset each other, leading to an insignificant response in  $v_d$ .

The Zhang parameterization, which includes stomatal response to VPD, captures the reductions in observed  $v_d$  and inferred  $g_{s,O3}$  at *Hyy* and *Blo* under high *VPD* conditions. Yet in addition it also predicts significant reduction in  $g_{ns,O3}$  at *Ha* due to low relative humidity, resulting in a large reduction in  $v_d$  not supported by the observations. The Wesely parameterization does capture the inferred responses of  $v_d$  and individual components at *Blo* within statistical uncertainty. At *Ha*, it predicts no changes in either  $g_{s,O3}$  or  $g_{ns,O3}$ , contradicting with our inference, but yields similar overall changes in  $v_d$ . In *Hyy* the responses are similar to those during extreme heat. We conclude that successfully predicting the reduction in  $g_{s,O3}$  does not necessarily guarantee accurate modeling of  $v_d$  during high *VPD* days, due to the difficulty of reproducing the response of apparent  $g_{nsO3}$ .





**Figure 1.** Differences in medians summer daytime (9a.m.–3p.m.)  $v_d$ ,  $g_{s,03}$ , and  $g_{ns,03}$  between anomalous and normal days derived from the evapotranspiration partitioning method proposed by Nelson et al. (2018). Error bars indicate 95% confidence interval (constructed following Bauer, 1972) for the differences in medians.

It has generally been proposed that heat and dryness leads to reduction in  $g_s$ , causing reduction in  $v_d$  and worse  $O_3$  air quality (Emberson et al., 2013; Huang et al., 2016; M. Lin et al., 2019). Less attention is given to the potential importance of responses in  $g_{ns,O3}$  under similar conditions. While we generally see the expected reduction in  $g_s$  under hot or high *VPD* conditions, there is a variable response in the apparent  $g_{ns,O3}$ . Consequently, the impact on overall  $v_d$  can vary. The increases in  $g_{ns,O3}$  inferred during hot conditions may partially enhance or offset the reduction in  $g_{s,O3}$  at *Hyy* (Figures 1 vs. S2 in Supporting Information S1), while dominating the overall response of  $v_d$  to anomalous conditions at *Blo* and *Ha*. Common big-leaf deposition models are unable to predict these responses in  $g_{ns,O3}$ , highlighting a need for better understanding the relationship between non-stomatal  $O_3$  uptake and anomalous conditions.

We examine the possible factors (Clifton et al., 2020) and their potential contributions to the inferred increase in  $g_{ns O3}$  during hot days:

- 1. During hot days, latent heat may mostly come from the cooler shaded leaves instead of the hotter sunlit leaves (e.g., He et al., 2018). The inferred  $g_{s,w}$  may be low-biased comparing to normal days (Text S2 in Supporting Information S1). This implies that both the decrease in  $g_s$  and increase  $g_{ns,O3}$  may be exaggerated during hot days. Yet if we accept the general ecophysiological theory that  $g_s$  decreases with heat anomalies, the increases in inferred  $g_{ns,O3}$  at *Blo* and *Ha* are qualitatively robust. On the other hand, this adds to the doubt of robustness of the inferred changes of  $g_{ns,O3}$  in *Hyy*, where the signal is small, and the carbon-based partitioning (Figure S2 in Supporting Information S1) suggests reductions instead of increases in  $g_{ns,O3}$ .
- 2. Using an experimentally-determined activation energy (30 kJ mol<sup>-1</sup>) (Cape et al., 2009) and assuming an O<sub>3</sub> cuticular conductance of 0.1 cm s<sup>-1</sup> during normal days representative of model estimates over dense forests (Clifton et al., 2019; Simpson et al., 2012), we estimate that the increase in cuticular uptake during hot days would contribute approximately 0.042, 0.025 and 0.030 cm s<sup>-1</sup> to total increases in  $g_{n_{5}O3}$  at *Hyy*, *Blo* and *Ha*, respectively. This is not enough to explain the inferred magnitude of increase in  $g_{n_{5}O3}$  over *Blo* and *Ha*.
- 3. Using a base emission of at most 3.5 ng N m<sup>-2</sup> s<sup>-1</sup> for ordinary days (Munger et al., 1996; Pilegaard et al., 2006; Wolfe et al., 2011), a standard temperature-dependence function for soil NO emission (Steinkamp & Lawrence, 2011), and the assumption that at most 0.8 mol of  $O_3$  is removed by each mole of soil NO emitted (Kurpius & Goldstein, 2003), we calculate that during the increase in soil NO during hot days translates to no more than 0.002 cm s<sup>-1</sup> increase in  $v_d$  over the three sites, which is negligible.

4. Temperature generally promotes emissions of BVOC (Guenther et al., 1995). As certain monoterpenes or sesquiterpenes can rapidly scavenge  $O_3$  (Atkinson & Arey, 2003; Yee et al., 2018), higher temperatures may promote the inferred non-stomatal  $O_3$  deposition through reactions with these BVOC. Though not directly verified by observations, this hypothesis is supported by our finding of larger increases in  $g_{ns,O3}$  at *Blo* and *Ha*, where previous work has argued for the influence of BVOC on  $O_3$  uptake (Clifton et al., 2019; Fares et al., 2010; Goldstein et al., 2004; Kurpius & Goldstein, 2003), and the contrast at *Hyy* where BVOC are not considered important sinks of  $O_3$  in general (Rannik et al., 2012; Zhou et al., 2017).

### 3.2. Soil Dryness Anomalies

We find less consistency in the response of  $v_{d^2} g_{s,O3}$  and  $g_{ns,O3}$  to dry soil days. Rather than being roughly equally distributed across different years (as is the case with high *T* and high *VPD* days), the driest soil days tend to be concentrated over prolonged episodes within particular years. Therefore, we analyze these dry soil episodes individually and use *GPP* observations to gauge the level of ecosystem stress. This helps elucidate how different levels of drought stress may affect O<sub>3</sub> deposition.

At *Hyy* the criteria of tenth percentile leads to identification of short and long dry soil episodes (<10 days in 2003, 2005, 2013; 16 days in 2009 and 11 days in 2010). In all cases, the impacts of soil dryness on *GPP* are relatively modest (-20% to -5%), and the range of mean daytime  $v_d$  across individual episodes is large (0.41–0.65 cm s<sup>-1</sup>). We find no consistent relationship between *SWC*, *GPP*, other meteorological variables and  $v_d$  over the five episodes (Table S3 in Supporting Information S1). This implies that the dry soil anomalies at *Hyy* may not all be significant enough to trigger consistent responses in O<sub>3</sub> dry deposition. With temperature and *VPD* conditions similar to other days (Table 1), the models likewise predict little change in  $v_d$ ,  $g_{s,O3}$  and  $g_{ns,O3}$  to dry soil anomalies here.

At *Ha*, the dry soil days mainly occur in late August of 1995 and early August of 1999, and stronger down-regulation of *GPP* is also observed (-25% in 1995 and -21% in 1999). The 2 years have very distinct mean daytime  $v_d$  during dry soil days (0.43 cm s<sup>-1</sup> in 1995 vs. 0.77 cm s<sup>-1</sup> in 1999), and this difference is explained by differences in the apparent  $g_{n_5,03}$  (-0.07 cm s<sup>-1</sup> in 1995 vs. +0.32 cm s<sup>-1</sup> in 1999 relative to average) rather than  $g_{s,03}$  (-0.19 cm s<sup>-1</sup> in 1995 vs. -0.20 cm s<sup>-1</sup> in 1999 relative to average). The 1999 episode has slightly higher temperature, *VPD* and lower *SWC* (Table S3 in Supporting Information S1). Given the uncertainty in EC-based *GPP* estimates in *Ha* (Wehr, Munger, et al., 2016), it is difficult to judge how different the ecosystem stress during these two episodes is. It might require highly specific ecosystem processes or events (e.g., Clifton et al., 2019; Urbanski et al., 2007), rather than simply relying on meteorological and *GPP* observations, to explain the huge difference in  $g_{n_5,03}$  between the two episodes. The Zhang parameterization partially responds to soil dryness by reducing  $g_{s,03}$ . The model predicts a reduction in average  $v_d$  comparable to observation when all dry days are considered, but it is not able to simulate the difference between the 1995 and 1999 episodes specifically. The Wesely parameterization, meanwhile, produces no significant response to soil dryness.

At *Blo*, all but one 39 dry soil days originate from the one single episode in August and September 2004. Strong concurrent reductions in mean daytime  $v_d$  (-0.27 cm s<sup>-1</sup>, -42%) are observed, and reductions in both  $g_{s,O3}$  (-0.17 cm s<sup>-1</sup>, -58%) and  $g_{ns,O3}$  (-0.15 cm s<sup>-1</sup>, -40%) are inferred. That summer was characterized by average *T* but extremely low spring rainfall, and the coincident decline in *GPP* (-65%), suggests that the ecosystem may have been under prolonged and severe drought stress. Neither the Wesely and Zhang parameterizations are able to capture the reduction in  $g_s$ , due to the lack of explicit dependence on *SWC*. However, we note that other  $v_d$  parameterizations with explicit  $g_s$  dependence on *SWC* (Centoni, 2017; Emberson et al., 2000; Meyers et al., 1998; Simpson et al., 2012; Valmartin et al., 2014) may likewise not produce the reduction in  $v_d$  due to simplistic representations of  $g_{ns,O3}$ . While monoterpene emissions in pine forests are generally a function of temperature and less related to ecosystem productivity due to storage (Bouvier-Brown et al., 2009), the strong reduction in photosynthetic capacity here may still have hampered the de novo emissions of monoterpene (Schurgers et al., 2009), reducing the inferred  $g_{ns,O3}$ .

## 4. Conclusions

We use three long-term  $O_3$  EC datasets to quantify the response of  $O_3$  dry deposition, and inferred stomatal and non-stomatal deposition, to heat and dry anomalies. Despite distinct environmental and ecological conditions, we generally find:

- 1. Inferred stomatal conductance is consistently reduced when the air or soil become extremely dry (high VPD or low SWC).
- 2. During hot days, especially when heat is not strong enough to suppress photosynthetic activity, inferred non-stomatal conductance tends to increase.
- 3. The magnitudes of changes in inferred stomatal and non-stomatal conductance during heat and dry anomalies are generally comparable.
- 4. Current big-leaf parameterizations tend to perform poorly compared to the observations partly because of their inability to reproduce the changes in apparent non-stomatal deposition.

The consistent reduction in inferred  $g_s$  during high VPD and low SWC days is expected from plant ecophysiological theory (Granier et al., 2007; Jarvis, 1976; Y. S. Lin et al., 2015; Medlyn et al., 2011). This response is sometimes reproduced by specific big-leaf dry deposition models if the influence of VPD is directly accounted for. In contrast, while previous literature has discussed the positive relationship between T and non-stomatal O<sub>3</sub> deposition (Fares et al., 2010; Kurpius & Goldstein, 2003), and the possibility of positive relationship between  $v_d$  and T when ozonolysis from BVOC is a major in-canopy O<sub>3</sub> sink (Wolfe et al., 2011), we explicitly show that  $g_{ns,O3}$  significantly increases during hot days. This behavior is not captured in the common big-leaf dry deposition models. Even more "advanced" big-leaf deposition models that consider how leaf wetness and relative humidity increases cuticular deposition (Clifton et al., 2020; Zhang et al., 2003), tend to instead predict reductions in  $g_{ns,O3}$ during hot days.

We propose faster thermal decomposition on dry cuticles and increased emissions of highly reactive BVOC as plausible mechanisms behind the high inferred  $g_{ns,O3}$  during hot days at the sites we considered. The uncertainty in leaf temperature, and the potential bias in inferring  $g_{s,w}$  due to the impossibility to distinguish between sunlit and shaded canopy under our framework, should also be considered when interpreting the changes in inferred stomatal and non-stomatal uptake.

On the other hand, we find less consistency in the responses of  $v_{d}$ ,  $g_{s,O3}$  and  $g_{ns,O3}$  to dry soil, which is a more direct indicator of water availability to plants. Taking *GPP* as a proxy of ecosystem stress status, we hypothesize that the varying intensity of soil dryness may have distinct impacts on O<sub>3</sub> deposition because of impacts on plant ecophysiology (Medrano et al., 2002), BVOC emissions (Niinemets, 2010), or both. Previous work has suggested that drier soils can generally increase soil O<sub>3</sub> deposition (Fares et al., 2014; Massad et al., 2019; Mészáros et al., 2009; Stella, Loubet, et al., 2011), but since we do not infer a consistent increase in  $g_{ns,O3}$  during dry soil days, such an effect may not be universally important in these particular ecosystems. Our definition of dry soil days allows us to examine the effects across a range of soil dryness, but selection based on closeness to soil wilting point in the future may yield more consistent insight across sites due to the direct ecophysiological relevance.

While we use the commonly observed responses of BVOC emissions to heat and drought stress to argue for potential role of BVOC ozonolysis in the response of  $g_{ns,O3}$  to heat and dry anomalies, it must be noted that stresses are also able to alter the composition of emitted BVOC, and, therefore potentially the total O<sub>3</sub> reactivity (Bonn et al., 2019; S. Li et al., 2017; Niinemets, 2010; Peñuelas & Staudt, 2010). This may play a role in the response of O<sub>3</sub> dry deposition during hot and dry anomalies, but the precise mechanisms remain largely unknown.

This work highlights the importance of changes in both stomatal and non-stomatal pathways in the response of  $O_3$  deposition during hot and dry anomalies, and the general inability of big-leaf parameterizations to reproduce the inferred responses in total  $v_{d}$ . This may lead to considerable error in predicting and attributing surface  $O_3$  changes during hot and dry episodes. We estimate the direct impacts on  $O_3$  to a first order using sensitivity simulations from GEOS-Chem (Figure 2) following the approach of Wong et al. (2019) (see Text S3 in Supporting Information S1), and find differences in  $O_3$  during heat and dry anomalies of up to 3–5 ppb that would not be correctly reproduced by the big leaf models. Modeling stomatal  $O_3$  uptake can be readily improved by applying more updated ecophysiological theories (Centoni, 2017; Lei et al., 2020; Valmartin et al., 2014), but our findings imply important limitations in our understanding of the environmental controls on non-stomatal  $O_3$  deposition. These





**Figure 2.** Estimated effect of observationally-inferred ("Inferred") and modeled ("Wesely" and "Zhang")  $v_d$  difference in anomalous days on surface ozone ( $\Delta O_3$ ) and its component attributable to stomatal ( $\Delta O_{3,s}$ ) and non-stomatal pathways ( $\Delta O_{3,ns}$ ). The sensitivity of surface  $O_3$  to  $v_d$  ( $\beta$ ) is derived from the simulations of Wong et al. (2019).

are best addressed with a combination of direct  $O_3$  flux and other concurrent measurements (e.g., soil moisture, BVOC speciation, canopy wetness) measurements (Clifton et al., 2020). Simultaneous monitoring of  $O_3$  and BVOC oxidation products fluxes may provide a novel and effective tool to study non-stomatal ozone deposition from in-canopy BVOC ozonolysis (Holzinger et al., 2005; Vermeuel et al., 2021).

Our work is an example that demonstrates the potential of applying data-driven techniques to help partition ET and infer  $g_s$ , which is crucial to partition the changes in  $v_d$  into stomatal and non-stomatal components. Data-driven techniques require only basic meteorological measurements in addition to EC fluxes, making them highly applicable to cross-site and long-term studies, providing a promising way to improve water vapor-based inference of  $g_s$ . Yet the discrepancies in estimated ET partitioning between different data-driven methods can be substantial (Nelson et al., 2020). Machine learning models might also be prone to overfitting, which we test in Table S4 in Supporting Information S1. The potential effects of overfitting on ET partitioning remains warrants future study, but we also find that our conclusions from this approach are consistent with other accepted portioning techniques (Text S1 in Supporting Information S1). Thus, the machine-learning technique could help alleviate, but does not fully eliminate, the limitations of water vapor-based inference of  $g_s$ . Further evidence from independent inferences of  $g_s$  by other means (e.g., sap flux, CO<sub>2</sub> flux, OCS flux (e.g., Wehr, Commane, et al., 2017)) will be particularly valuable.

Given the functional diversity of plants, the scarcity of observation (O. E. Clifton et al., 2020), and the importance of the spatiotemporal dynamics of dry deposition on understanding and prediction of surface  $O_3$  (Baublitz et al., 2020; Clifton et al., 2020; M. Lin et al., 2019, 2017; Wong et al., 2019), and vegetation impacts (Mills et al., 2011; Ronan et al., 2020), direct  $O_3$  flux observations must be expanded in both space and time to deepen our understanding of surface  $O_3$  concentrations with global change. Longer observational datasets with greater spatial density have the added benefit of potentially allowing big-data type approaches to model the complex phenomenon of  $O_3$  deposition (Silva et al., 2019).

#### **Data Availability Statement**

Ozone flux measurements from Harvard Forest is available through J. W. Munger et al. (1996), and retrieved from Harvard Forest Data Archive. Ozone flux measurements from Hyytiälä Forest is available through Keronen et al. (2003), and retrieved from SmartSMEAR. Ozone flux measurements from Blodgett Forest is available

from Fares et al. (2010). Additional site characteristics is obtained from biomass inventory data (HF069) of Harvard Forest Archive for Harvard (W. Munger and Wofsy, 2021) and Launiainen et al. (2016) for Hyytiälä. The processed data directly used in this analysisd; is freely available in the Boston University Institutional Repository, open BU (https://open.bu.edu/handle/2144/43002).

## References

Ainsworth, E. A., Yendrek, C. R., Sitch, S., Collins, W. J., & Emberson, L. D. (2012). The effects of tropospheric ozone on net primary productivity and implications for climate change. *Annual Review of Plant Biology*, 63(1), 637–661. https://doi.org/10.1146/annurev-arplant-042110-103829 Alduchov, O. A., & Eskridge, R. E. (1996). Improved Magnus form approximation of saturation vapor pressure. *Journal of Applied Meteorology*, 35(4), 601–609. https://doi.org/10.1175/1520-0450(1996)035<0601;imfaos>2.0.co;2

- Atkinson, R., & Arey, J. (2003). Gas-phase tropospheric chemistry of biogenic volatile organic compounds: A review. Atmospheric Environment, 37, 197–219. https://doi.org/10.1016/S1352-2310(03)00391-1
- Baublitz, C. B., Fiore, A. M., Clifton, O. E., Mao, J., Li, J., Correa, G., et al. (2020). Sensitivity of tropospheric ozone over the Southeast USA to dry deposition. *Geophysical Research Letters*, 47(7), 1–10. https://doi.org/10.1029/2020gl087158
- Bauer, D. F. (1972). Constructing confidence sets using rank statistics. Journal of the American Statistical Association, 67(339), 687–690. https:// doi.org/10.1080/01621459.1972.10481279
- Bonn, B., Magh, R. K., Rombach, J., & Kreuzwieser, J. (2019). Biogenic isoprenoid emissions under drought stress: Different responses for isoprene and terpenes. *Biogeosciences*. https://doi.org/10.5194/bg-16-4627-2019
- Bouvier-Brown, N. C., Holzinger, R., Palitzsch, K., & Goldstein, A. H. (2009). Large emissions of sesquiterpenes and methyl chavicol quantified from branch enclosure measurements. Atmospheric Environment, 43(2), 389–401. https://doi.org/10.1016/j.atmosenv.2008.08.039
  - Cape, J. N., Hamilton, R., & Heal, M. R. (2009). Reactive uptake of ozone at simulated leaf surfaces: Implications for "non-stomatal" ozone flux. *Atmospheric Environment*. https://doi.org/10.1016/j.atmosenv.2008.11.007
  - Centoni, F. (2017). Global scale modelling of ozone deposition processes and interaction between surface ozone and climate change A thesis presented for the degree. The University of Edinburgh. University of Edinburgh.
  - Chu, H., Baldocchi, D. D., John, R., Wolf, S., & Reichstein, M. (2017). Fluxes all of the time? A primer on the temporal representativeness of FLUXNET. Journal of Geophysical Research: Biogeosciences. https://doi.org/10.1002/2016JG003576
- Clifton, O. E., Fiore, A. M., Munger, J. W., Malyshev, S., Horowitz, L. W., Shevliakova, E., et al. (2017). Interannual variability in ozone removal by a temperate deciduous forest. *Geophysical Research Letters*, 44(1), 542–552. https://doi.org/10.1002/2016GL070923
- Clifton, O. E., Fiore, A. M., Munger, J. W., & Wehr, R. (2019). Spatiotemporal controls on observed daytime ozone deposition velocity over northeastern U.S. Forests during summer. *Journal of Geophysical Research: Atmospheres*, 124(10), 5612–5628. https://doi. org/10.1029/2018JD029073
- Clifton, O. E., Paulot, F., Fiore, A. M., Horowitz, L. W., Correa, G., Baublitz, C. B., et al. (2020). Influence of dynamic ozone dry deposition on ozone pollution. *Journal of Geophysical Research: Atmospheres, 125*(8), e2020JD032398. https://doi.org/10.1029/2020jd032398
- Dai, A., & Zhao, T. (2017). Uncertainties in historical changes and future projections of drought. Part I: Estimates of historical drought changes. *Climatic Change*. https://doi.org/10.1007/s10584-016-1705-2
- Emberson, L. D., Kitwiroon, N., Beevers, S., Büker, P., & Cinderby, S. (2013). Scorched Earth: How will changes in the strength of the vegetation sink to ozone deposition affect human health and ecosystems? *Atmospheric Chemistry and Physics*, 13(14), 6741–6755. https://doi. org/10.5194/acp-13-6741-2013
- Emberson, L. D., Wieser, G., & Ashmore, M. R. (2000). Modelling of stomatal conductance and ozone flux of Norway spruce: Comparison with field data. *Environmental Pollution*, 109, 393–402. https://doi.org/10.1016/S0269-7491(00)00042-7
- Fares, S., McKay, M., Holzinger, R., & Goldstein, A. H. (2010). Ozone fluxes in a *Pinus ponderosa* ecosystem are dominated by non-stomatal processes: Evidence from long-term continuous measurements. *Agricultural and Forest Meteorology*, 150(3), 420–431. https://doi. org/10.1016/j.agrformet.2010.01.007
- Fares, S., Savi, F., Muller, J., Matteucci, G., & Paoletti, E. (2014). Simultaneous measurements of above and below canopy ozone fluxes help partitioning ozone deposition between its various sinks in a Mediterranean Oak Forest. Agricultural and Forest Meteorology, 198, 181–191. https://doi.org/10.1016/j.agrformet.2014.08.014
- Fares, S., Weber, R., Park, J. H., Gentner, D., Karlik, J., & Goldstein, A. H. (2012). Ozone deposition to an orange orchard: Partitioning between stomatal and non-stomatal sinks. *Environmental Pollution*, 169, 258–266. https://doi.org/10.1016/j.envpol.2012.01.030
- Filleul, L., Cassadou, S., Médina, S., Fabres, P., Lefranc, A., Eilstein, D., et al. (2006). The relation between temperature, ozone, and mortality in nine French cities during the heat wave of 2003. Environmental Health Perspectives. https://doi.org/10.1289/ehp.8328
- Foken, T. (2006). 50 years of the Monin-Obukhov similarity theory. Boundary-Layer Meteorology. https://doi.org/10.1007/s10546-006-9048-6
  Fowler, D., Pilegaard, K., Sutton, M. A., Ambus, P., Raivonen, M., Duyzer, J., et al. (2009). Atmospheric composition change: Ecosystems-Atmosphere interactions. Atmospheric Environment. https://doi.org/10.1016/j.atmosenv.2009.07.068
- Freire, L. S., Gerken, T., Ruiz-Plancarte, J., Wei, D., Fuentes, J. D., Katul, G. G., et al. (2017). Turbulent mixing and removal of ozone within an Amazon rainforest canopy. *Journal of Geophysical Research*, 122(5), 2791–2811. https://doi.org/10.1002/2016JD026009
- Gerosa, G., Derghi, F., & Cieslik, S. (2007). Comparison of different algorithms for stomatal ozone flux determination from micrometeorological measurements. Water, Air, and Soil Pollution, 179(1–4), 309–321. https://doi.org/10.1007/s11270-006-9234-7
- Gerosa, G., Vitale, M., Finco, A., Manes, F., Denti, A. B., & Cieslik, S. (2005). Ozone uptake by an evergreen Mediterranean Forest (Quercus ilex) in Italy. Part I: Micrometeorological flux measurements and flux partitioning. *Atmospheric Environment*, 39(18), 3255–3266. https://doi.org/10.1016/j.atmosenv.2005.01.056
- Goldstein, A. H., McKay, M., Kurpius, M. R., Schade, G. W., Lee, A., Holzinger, R., & Rasmussen, R. A. (2004). Forest thinning experiment confirms ozone deposition to forest canopy is dominated by reaction with biogenic VOCs. *Geophysical Research Letters*. https://doi.org/10.1029/2004GL021259
- Granier, A., Reichstein, M., Bréda, N., Janssens, I. A., Falge, E., Ciais, P., et al. (2007). Evidence for soil water control on carbon and water dynamics in European forests during the extremely dry year: 2003. Agricultural and Forest Meteorology. https://doi.org/10.1016/j. agrformet.2006.12.004
- Guenther, A. B., Hewitt, C. N., Erickson, D., Fall, R., Geron, C., Graedel, T., et al. (1995). A global model of natural volatile organic compound emissions. *Journal of Geophysical Research*, 100(D5), 8873–8892. https://doi.org/10.1029/94JD02950

#### Acknowledgments

A. Y. H. Wong and J. A. Geddes are supported by an NSF CAREER grant (ATM-1750328). IM thanks ACCC Flagship funded by the academy of Finland Grant No. 337549. J. A. Ducker and C. D. Holmes were supported by NSF Grant 1848372. Harvard Forest observations were supported in part by the U.S. Department of Energy, Office of Science (BER), and NSF Long Term Ecological Research. This work used eddy covariance data acquired and shared by the FLUXNET community, including these networks: AmeriFlux, AfriFlux, AsiaFlux, CarboAfrica, CarboEuropeIP, CarboItaly, CarboMont, ChinaFlux, Fluxnet-Canada, GreenGrass, ICOS, KoFlux, LBA, NECC, OzFlux-TERN, TCOS-Siberia, and USCCC. The FLUXNET eddy covariance data processing and harmonization (Pastorello et al., 2020) was carried out by the ICOS Ecosystem Thematic Center, AmeriFlux Management Project and Fluxdata project of FLUXNET, with the support of CDIAC, and the OzFlux, ChinaFlux and AsiaFlux offices.

- Hardacre, C., Wild, O., & Emberson, L. (2015). An evaluation of ozone dry deposition in global scale chemistry climate models. Atmospheric Chemistry and Physics, 15(11), 6419–6436. https://doi.org/10.5194/acp-15-6419-2015
- He, L., Chen, J. M., Gonsamo, A., Luo, X., Wang, R., Liu, Y., & Liu, R. (2018). Changes in the shadow: The shifting role of shaded leaves in global carbon and water cycles under climate change. *Geophysical Research Letters*, 45(10), 5052–5061. https://doi.org/10.1029/2018GL077560
- Hogg, A., Uddling, J., Ellsworth, D., Carroll, M. A., Pressley, S., Lamb, B., & Vogel, C. (2007). Stomatal and non-stomatal fluxes of ozone to a northern mixed hardwood forest. *Tellus B: Chemical and Physical Meteorology*, https://doi.org/10.3402/tellusb.v59i3.17027
- Holzinger, R., Lee, A., Paw U, K. T., & Goldstein, A. H. (2005). Observations of oxidation products above a forest imply biogenic emissions of very reactive compounds. *Atmospheric Chemistry and Physics*, 5(1), 67–75. https://doi.org/10.5194/acp-5-67-2005
- Huang, L., McDonald-Buller, E. C., McGaughey, G., Kimura, Y., & Allen, D. T. (2016). The impact of drought on ozone dry deposition over eastern Texas. Atmospheric Environment, 127, 176–186. https://doi.org/10.1016/j.atmosenv.2015.12.022
- Jarvis, P. G. (1976). The interpretation of the variations in leaf water potential and stomatal conductance found in Canopies in the field. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 273(927), 593–610. https://doi.org/10.1098/rstb.1976.0035
- Jerrett, M., Burnett, R. T., Arden Pope, C., Ito, K., Thurston, G., Krewski, D., et al. (2009). Long-term ozone exposure and mortality. New England Journal of Medicine, 360(11), 1085–1095. https://doi.org/10.1056/NEJMoa0803894
- Keronen, P., Reissell, A., Rannik, Ü., Pohja, T., Siivola, E., Hiltunen, V., et al. (2003). Ozone flux measurements over a Scots pine forest using eddy covariance method: Performance evaluation and comparison with flux-profile method. *Boreal Environment Research*, 8(4), 425–443. Retrieved from http://www.scopus.com/inward/record.url?eid=2-s2.0-0347884158&partnerID=40&md5=4ad114fb52c557d36cc8a0ec1ab8bb7e
- Kurpius, M. R., & Goldstein, A. H. (2003). Gas-phase chemistry dominates O<sub>3</sub> loss to a forest, implying a source of aerosols and hydroxyl radicals to the atmosphere. *Geophysical Research Letters*. https://doi.org/10.1029/2002GL016785
- Launiainen, S., Katul, G. G., Kolari, P., Lindroth, A., Lohila, A., Aurela, M., et al. (2016). Do the energy fluxes and surface conductance of boreal coniferous forests in Europe scale with leaf area? *Global Change Biology*. https://doi.org/10.1111/gcb.13497
- Lei, Y., Yue, X., Liao, H., Gong, C., & Zhang, L. (2020). Implementation of Yale interactive terrestrial biosphere model v1.0 into GEOS-Chem v12.0.0: A tool for biosphere-chemistry interactions. Geoscientific Model Development. https://doi.org/10.5194/gmd-13-1137-2020
- Li, Q., Gabay, M., Rubin, Y., Raveh-Rubin, S., Rohatyn, S., Tatarinov, F., et al. (2019). Investigation of ozone deposition to vegetation under warm and dry conditions near the Eastern Mediterranean coast. Science of the Total Environment. https://doi.org/10.1016/j.scitotenv.2018.12.272
- Li, S., Harley, P. C., & Niinemets, Ü. (2017). Ozone-induced foliar damage and release of stress volatiles is highly dependent on stomatal openness and priming by low-level ozone exposure in Phaseolus vulgaris. *Plant, Cell and Environment*, 40(9), 1984–2003. https://doi.org/10.1111/ pce.13003
- Lin, M., Horowitz, L. W., Payton, R., Fiore, A. M., & Tonnesen, G. (2017). US surface ozone trends and extremes from 1980 to 2014: Quantifying the roles of rising Asian emissions, domestic controls, wildfires, and climate. Atmospheric Chemistry and Physics, 17(4), 2943–2970. https:// doi.org/10.5194/acp-17-2943-2017
- Lin, M., Malyshev, S., Shevliakova, E., Paulot, F., Horowitz, L. W., Fares, S., et al. (2019). Sensitivity of ozone dry deposition to ecosystem-atmosphere interactions: A critical appraisal of observations and simulations. *Global Biogeochemical Cycles*, 33. https://doi. org/10.1029/2018gb006157
- Lin, Y. S., Medlyn, B. E., Duursma, R. A., Prentice, I. C., Wang, H., Baig, S., et al. (2015). Optimal stomatal behaviour around the world. Nature Climate Change, 5(5), 459–464. https://doi.org/10.1038/nclimate2550
- Mammarella, I., Kolari, P., Rinne, J., Keronen, P., Pumpanen, J., & Vesala, T. (2007). Determining the contribution of vertical advection to the net ecosystem exchange at Hyytiäläforest, Finland. *Tellus Series B Chemical and Physical Meteorology*. https://doi. org/10.1111/j.1600-0889.2007.00306.x
- Massad, R. S., Lathière, J., Strada, S., Perrin, M., Personne, E., Stéfanon, M., et al. (2019). Reviews and syntheses: Influences of landscape structure and land uses on local to regional climate and air quality. *Biogeosciences*, 16(11), 2369–2408. https://doi.org/10.5194/bg-16-2369-2019
- Medlyn, B. E., Duursma, R. A., Eamus, D., Ellsworth, D. S., Prentice, I. C., Barton, C. V. M., et al. (2011). Reconciling the optimal and empirical approaches to modelling stomatal conductance. *Global Change Biology*, 17(6), 2134–2144. https://doi.org/10.1111/j.1365-2486.2010.02375.x
- Medrano, H., Escalona, J. M., Bota, J., Gulías, J., & Flexas, J. (2002). Regulation of photosynthesis of C3 plants in response to progressive drought: Stomatal conductance as a reference parameter. Annals of Botany. https://doi.org/10.1093/aob/mcf079
- Meehl, G. A., & Tebaldi, C. (2004). More intense, more frequent, and longer lasting heat waves in the 21st century. *Science*. https://doi.org/10.1126/science.1098704
- Mészáros, R., Horváth, L., Weidinger, T., Neftel, A., Nemitz, E., Dämmgen, U., et al. (2009). Measurement and modelling ozone fluxes over a cut and fertilized grassland. *Biogeosciences*, 6(10), 1987–1999. https://doi.org/10.5194/bg-6-1987-2009
- Meyers, T. P., Finkelstein, P., Clarke, J., Ellestad, T. G., & Sims, P. F. (1998). A multilayer model for inferring dry deposition using standard meteorological measurements. *Journal of Geophysical Research*, 103(D17), 22645–22661. https://doi.org/10.1029/98JD01564
- Mills, G., Hayes, F., Simpson, D., Emberson, L., Norris, D., Harmens, H., & Büker, P. (2011). Evidence of widespread effects of ozone on crops and (semi-)natural vegetation in Europe (1990–2006) in relation to AOT40- and flux-based risk maps. *Global Change Biology*, 17(1), 592–613. https://doi.org/10.1111/j.1365-2486.2010.02217.x
- Monin, A. S., & Obukhov, A. M. (1954). Basic laws of turbulent mixing in the surface layer of the atmosphere. Contribution of Geophysical Institute Academy of Sciences of the USSR, 24(151), 163–187.
- Monteith, J. L. (1965). Evaporation and environment. Symposia of the Society for Experimental Biology, 19, 205–234.
- Munger, J. W., Wofsy, S. C., Bakwin, P. S., Fan, S.-M., Goulden, M. L., Daube, B. C., et al. (1996). Atmospheric deposition of reactive nitrogen oxides and ozone in a temperate deciduous forest and a subarctic woodland: 1. Measurements and mechanisms. *Journal of Geophysical Research*, 101(D7), 12639–12657. https://doi.org/10.1029/96jd00230
- Nelson, J. A., Carvalhais, N., Cuntz, M., Delpierre, N., Knauer, J., Ogée, J., et al. (2018). Coupling water and carbon fluxes to constrain estimates of transpiration: The TEA algorithm. *Journal of Geophysical Research: Biogeosciences*. https://doi.org/10.1029/2018JG004727
- Nelson, J. A., Pérez-Priego, O., Zhou, S., Poyatos, R., Zhang, Y., Blanken, P. D., et al. (2020). Ecosystem transpiration and evaporation: Insights from three water flux partitioning methods across FLUXNET sites. *Global Change Biology*, 26(12), 6916–6930. https://doi.org/10.1111/ gcb.15314
- Niinemets, Ü. (2010). Mild versus severe stress and BVOCs: Thresholds, priming and consequences. Trends in Plant Science. https://doi. org/10.1016/j.tplants.2009.11.008
- Pastorello, G., Trotta, C., Canfora, E., Chu, H., Christianson, D., Cheah, Y. W., et al. (2020). The FLUXNET2015 dataset and the ONEFlux processing pipeline for eddy covariance data. *Scientific Data*, 7(1), 225. https://doi.org/10.1038/s41597-020-0534-3

Peñuelas, J., & Staudt, M. (2010). BVOCs and global change. *Trends in Plant Science*. https://doi.org/10.1016/j.tplants.2009.12.005 Perkins, S. E., & Alexander, L. V. (2013). On the measurement of heat waves. *Journal of Climate*. https://doi.org/10.1175/JCLI-D-12-00383.1

- Perkins, S. E., Alexander, L. V., & Nairn, J. R. (2012). Increasing frequency, intensity and duration of observed global heatwaves and warm spells. *Geophysical Research Letters*. https://doi.org/10.1029/2012GL053361
- Pilegaard, K., Skiba, U., Ambus, P., Beier, C., Brüggemann, N., Butterbach-Bahl, K., et al. (2006). Factors controlling regional differences in forest soil emission of nitrogen oxides (NO and N<sub>2</sub>O). *Biogeosciences*. https://doi.org/10.5194/bg-3-651-2006
- Pleim, J., & Ran, L. (2011). Surface flux modeling for air quality applications. Atmosphere. https://doi.org/10.3390/atmos2030271
  - Rannik, Ü., Altimir, N., Mammarella, I., Bäck, J., Rinne, J., Ruuskanen, T. M., et al. (2012). Ozone deposition into a boreal forest over a decade of observations: Evaluating deposition partitioning and driving variables. Atmospheric Chemistry and Physics, 12(24), 12165–12182. https:// doi.org/10.5194/acp-12-12165-2012
  - Ronan, A. C., Ducker, J. A., Schnell, J. L., & Holmes, C. D. (2020). Have improvements in ozone air quality reduced ozone uptake into plants? *Elementa*, 8(1). https://doi.org/10.1525/elementa.399
  - Samaniego, L., Thober, S., Kumar, R., Wanders, N., Rakovec, O., Pan, M., et al. (2018). Anthropogenic warming exacerbates European soil moisture droughts. *Nature Climate Change*. https://doi.org/10.1038/s41558-018-0138-5
  - Schurgers, G., Arneth, A., Holzinger, R., & Goldstein, A. H. (2009). Process-based modelling of biogenic monoterpene emissions combining production and release from storage. Atmospheric Chemistry and Physics. https://doi.org/10.5194/acp-9-3409-2009
  - Silva, S. J., Heald, C. L., Ravela, S., Mammarella, I., & Munger, J. W. (2019). A deep learning parameterization for ozone dry deposition velocities. *Geophysical Research Letters*, 46(2), 983–989. https://doi.org/10.1029/2018GL081049
  - Simpson, D., Benedictow, A., Berge, H., Bergström, R., Emberson, L. D., Fagerli, H., et al. (2012). The EMEP MSC-W chemical transport model—Technical description. Atmospheric Chemistry and Physics, 12(16), 7825–7865. https://doi.org/10.5194/acp-12-7825-2012
  - Steinkamp, J., & Lawrence, M. G. (2011). Improvement and evaluation of simulated global biogenic soil NO emissions in an AC-GCM. Atmospheric Chemistry and Physics, 11(12), 6063–6082. https://doi.org/10.5194/acp-11-6063-2011
  - Stella, P., Loubet, B., de Berranger, C., Charrier, X., Ceschia, E., Gerosa, G., et al. (2019). Soil ozone deposition: Dependence of soil resistance to soil texture. *Atmospheric Environment*. https://doi.org/10.1016/j.atmosenv.2018.11.036
  - Stella, P., Loubet, B., Lamaud, E., Laville, P., & Cellier, P. (2011). Ozone deposition onto bare soil: A new parameterisation. Agricultural and Forest Meteorology, 151(6), 669–681. https://doi.org/10.1016/j.agrformet.2011.01.015
  - Stella, P., Personne, E., Loubet, B., Lamaud, E., Ceschia, E., Béziat, P., et al. (2011). Predicting and partitioning ozone fluxes to maize crops from sowing to harvest: The Surfatm-O3 model. *Biogeosciences*, 8(10), 2869–2886. https://doi.org/10.5194/bg-8-2869-2011
  - Sun, S., Moravek, A., Trebs, I., Kesselmeier, J., & Sörgel, M. (2016). Investigation of the influence of liquid surface films on O3 and PAN deposition to plant leaves coated with organic/inorganic solution. *Journal of Geophysical Research*, 121(23), 14239–14256. https://doi. org/10.1002/2016JD025519
  - Tai, A. P. K., Martin, M. V., & Heald, C. L. (2014). Threat to future global food security from climate change and ozone air pollution. Nature Climate Change, 4(9), 817–821. https://doi.org/10.1038/nclimate2317
  - Urbanski, S., Barford, C., Wofsy, S., Kucharik, C., Pyle, E., Budney, J., et al. (2007). Factors controlling CO<sub>2</sub> exchange on timescales from hourly to decadal at Harvard Forest. *Journal of Geophysical Research*. https://doi.org/10.1029/2006JG000293
  - Valmartin, M., Heald, C. L., & Arnold, S. R. (2014). Coupling dry deposition to vegetation phenology in the Community Earth System Model: Implications for the simulation of surface O<sub>3</sub>. *Geophysical Research Letters*, 41(8), 2988–2996. https://doi.org/10.1002/2014GL059651
  - Vermeuel, M. P., Cleary, P. A., Desai, A. R., & Bertram, T. H. (2021). Simultaneous measurements of O<sub>3</sub> and HCOOH vertical fluxes indicate rapid in-Canopy Terpene chemistry enhances O<sub>3</sub> removal over mixed temperate forests. *Geophysical Research Letters*. https://doi. org/10.1029/2020gl090996
  - Wehr, R., Commane, R., Munger, J. W., Barry Mcmanus, J., Nelson, D. D., Zahniser, M. S., et al. (2017). Dynamics of canopy stomatal conductance, transpiration, and evaporation in a temperate deciduous forest, validated by carbonyl sulfide uptake. *Biogeosciences*, 14(2), 389–401. https://doi.org/10.5194/bg-14-389-2017
  - Wehr, R., Munger, J. W., McManus, J. B., Nelson, D. D., Zahniser, M. S., Davidson, E. A., et al. (2016). Seasonality of temperate forest photosynthesis and daytime respiration. *Nature*, 534(7609), 680–683. https://doi.org/10.1038/nature17966
  - Wesely, M. L. (1989). Parameterization of surface resistances to gaseous dry deposition in regional-scale numerical models. Atmospheric Environment, 41(SUPPL), 52–63. https://doi.org/10.1016/j.atmosenv.2007.10.058
  - Wesely, M. L., & Hicks, B. B. (1977). Some factors that affect the deposition rates of sulfur dioxide and similar gases on vegetation. Journal of the Air Pollution Control Association, 27(11), 1110–1116. https://doi.org/10.1080/00022470.1977.10470534
  - Wesely, M. L., & Hicks, B. B. (2000). A review of the current status of knowledge on dry deposition. Atmospheric Environment, 34(12–14), 2261–2282. https://doi.org/10.1016/S1352-2310(99)00467-7
  - Wilcoxon, F. (1945). Individual comparisons by ranking methods. Biometrics Bulletin. https://doi.org/10.2307/3001968
  - Wild, O. (2007). Modelling the global tropospheric ozone budget: Exploring the variability in current models. *Atmospheric Chemistry and Physics*, 7(10), 2643–2660. https://doi.org/10.5194/acp-7-2643-2007
  - Wittig, V. E., Ainsworth, E. A., Naidu, S. L., Karnosky, D. F., & Long, S. P. (2009). Quantifying the impact of current and future tropospheric ozone on tree biomass, growth, physiology and biochemistry: A quantitative meta-analysis. *Global Change Biology*, 15(2), 396–424. https:// doi.org/10.1111/j.1365-2486.2008.01774.x
  - Wolfe, G. M., Thornton, J. A., McKay, M., & Goldstein, A. H. (2011). Forest-atmosphere exchange of ozone: Sensitivity to very reactive biogenic VOC emissions and implications for in-canopy photochemistry. *Atmospheric Chemistry and Physics*. https://doi.org/10.5194/ acp-11-7875-2011
  - Wong, A. Y. H., Geddes, J. A., Tai, A. P. K., & Silva, S. J. (2019). Importance of dry deposition parameterization choice in global simulations of surface ozone. Atmospheric Chemistry and Physics, 19(22), 14365–14385. https://doi.org/10.5194/acp-19-14365-2019
  - Yee, L. D., Isaacman-VanWertz, G., Wernis, R. A., Meng, M., Rivera, V., Kreisberg, N. M., et al. (2018). Observations of sesquiterpenes and their oxidation products in central Amazonia during the wet and dry seasons. *Atmospheric Chemistry and Physics*. https://doi.org/10.5194/ acp-18-10433-2018
  - Zhang, L., Brook, J. R., & Vet, R. (2002). On ozone dry deposition—with emphasis on non-stomatal uptake and wet canopies. Atmospheric Environment, 36(30), 4787–4799. https://doi.org/10.1016/S1352-2310(02)00567-8
  - Zhang, L., Brook, J. R., & Vet, R. (2003). A revised parameterization for gaseous dry deposition in air-quality models. *Atmospheric Chemistry* and *Physics*, 3(6), 2067–2082. https://doi.org/10.5194/acp-3-2067-2003
  - Zhou, P., Ganzeveld, L., Rannik, U., Zhou, L., Gierens, R., Taipale, D., et al. (2017). Simulating ozone dry deposition at a boreal forest with a multi-layer canopy deposition model. *Atmospheric Chemistry and Physics*, 17(2), 1361–1379. https://doi.org/10.5194/acp-17-1361-2017
  - Zona, D., Gioli, B., Fares, S., De Groote, T., Pilegaard, K., Ibrom, A., & Ceulemans, R. (2014). Environmental controls on ozone fluxes in a poplar plantation in Western Europe. *Environmental Pollution*, 184, 201–210. https://doi.org/10.1016/j.envpol.2013.08.032

Zscheischler, J., Reichstein, M., Harmeling, S., Rammig, A., Tomelleri, E., & Mahecha, M. D. (2014). Extreme events in gross primary production: A characterization across continents. *Biogeosciences*. https://doi.org/10.5194/bg-11-2909-2014

# **References From the Supporting Information**

Breiman, L. (2001). Random forrests. *Machine Learning*, 45(1), 5–32.

- Clifton, O. E., Fiore, A. M., Massman, W. J., Baublitz, C. B., Coyle, M., Emberson, L., et al. (2020). Dry deposition of ozone over land: Processes, measurement, and modeling. *Reviews of Geophysics*, 58(1). https://doi.org/10.1029/2019RG000670
- Ducker, J. A., Holmes, C. D., Keenan, T. F., Fares, S., Goldstein, A. H., Mammarella, I., et al. (2018). Synthetic ozone deposition and stomatal uptake at flux tower sites. *Biogeosciences*, 15(17), 5395–5413. https://doi.org/10.5194/bg-15-5395-2018
- Munger, W., & Wofsy, S. (2021). Biomass inventories at Harvard Forest EMS Tower since 1993. Harvard Forest Data Archive: HF069 (v.36). Environmental Data Initiative. https://doi.org/10.6073/pasta/5c2f17c295413da2a2a091fd7696af40
- Franks, P. J., Bonan, G. B., Berry, J. A., Lombardozzi, D. L., Holbrook, N. M., Herold, N., & Oleson, K. W. (2018). Comparing optimal and empirical stomatal conductance models for application in Earth system models. *Global Change Biology*, 24(12), 5708–5723. https://doi. org/10.1111/gcb.14445
- Hu, X., & Lei, H. (2021). Evapotranspiration partitioning and its interannual variability over a winter wheat-summer maize rotation system in the North China Plain. Agricultural and Forest Meteorology, 310, 108635. https://doi.org/10.1016/j.agrformet.2021.108635
- Koenker, R., & Machado, J. A. F. (1999). Goodness of fit and related inference processes for quantile regression. Journal of the American Statistical Association, 94(448), 1296–1310. https://doi.org/10.1080/01621459.1999.10473882

Meinshausen, N. (2006). Quantile regression forests. Journal of Machine Learning Research, 7, 983-999.

- Stoy, P. C., El-Madany, T. S., Fisher, J. B., Gentine, P., Gerken, T., Good, S. P., et al. (2019). Reviews and syntheses: Turning the challenges of partitioning ecosystem evaporation and transpiration into opportunities. *Biogeosciences*. https://doi.org/10.5194/bg-16-3747-2019
- Wang, Y. P., & Leuning, R. (1998). A two-leaf model for canopy conductance, photosynthesis and partitioning of available energy I: Model description and comparison with a multi-layered model. Agricultural and Forest Meteorology, 91(1–2), 89–111. https://doi.org/10.1016/ S0168-1923(98)00061-6
- Wang, Y., Jacob, D. J., & Logan, J. A. (1998). Global simulation of tropospheric O<sub>3</sub>-NO x-hydrocarbon chemistry—1. Model formulation. *Journal of Geophysical Research D: Atmospheres*, 103(3339), 10713–10725. https://doi.org/10.1029/98jd00158